Author's response to reviews

Title: The Rehabilitation Enhancing Aging Through Connected Health (REACH) Study: study protocol for a quasi-experimental clinical trial

Authors:

Meng Ni (nimeng8815@gmail.com)
Lorna Brown (lbrown26@partners.org)
Danielle Lawler (dlawler@partners.org)
Terry Ellis (tellis@bu.edu)
Tamara Deangelis (trork@bu.edu)
Nancy Latham (nlatham@bu.edu)
Jennifer Perloff (perloff@brandeis.edu)
Steve J. A Atlas (satlas@mgh.harvard.edu)
Sanja Percac-Lima (spercaclima@partners.org)
Jonathan Bean (Jonathan.Bean4@va.gov)

Version: 1 Date: 06 Sep 2017

Author’s response to reviews:

Dear editorial office,

The co-authors and I really appreciated your review! We have already made the revisions accordingly, including the abbreviations, declaration (Ethics approval and consent to participate, Consent for publication, Availability of data and materials, and Acknowledgement). Please refer to the newly uploaded manuscript with highlighted parts for the revised sections. Also, the revised manuscript without highlights is uploaded.

Please don’t hesitate to contact me if you have any further questions or concerns.

Sincerely,

Meng